

Since January 2020 Elsevier has created a COVID-19 resource centre with free information in English and Mandarin on the novel coronavirus COVID-19. The COVID-19 resource centre is hosted on Elsevier Connect, the company's public news and information website.

Elsevier hereby grants permission to make all its COVID-19-related research that is available on the COVID-19 resource centre - including this research content - immediately available in PubMed Central and other publicly funded repositories, such as the WHO COVID database with rights for unrestricted research re-use and analyses in any form or by any means with acknowledgement of the original source. These permissions are granted for free by Elsevier for as long as the COVID-19 resource centre remains active.



Contents lists available at ScienceDirect

Cities



journal homepage: www.elsevier.com/locate/cities

The COVID-19 pandemic and the livelihood of a vulnerable population: Evidence from women street vendors in urban Vietnam

Check for updates

Pham Tien Thanh^a, Pham Bao Duong^{b,*}

^a Faculty of Business Administration, Ton Duc Thang University, Ho Chi Minh City, Vietnam
^b Faculty of Economics and Rural Development, Vietnam National University of Agriculture, Hanoi, Vietnam

ARTICLE INFO

JEL classification: 131 138 Keywords: Economic burden COVID-19 Mitigation mechanisms Women street vendors Vulnerability

ABSTRACT

The COVID-19 pandemic has caused enormous detrimental impacts on a global scale. Street vendors are one of the most heavily affected groups since they lack the skills and resources to overcome shocks. This study examines the economic burden facing this group during the pandemic and their coping strategies and mitigation mechanisms in response to these adverse effects. We utilized a mixed-methods approach, wherein 91 women vendors completed a survey questionnaire and 15 women vendors were interviewed. These vendors were found to experience a large reduction in business and consumption. The businesses of immigrant vendors suffered more adverse effects than those of local vendors. Also, the vendors selling in wet market areas incurred greater economic burdens than those selling near schools or recreation centers. The vendors lacked coping strategies to sustain their businesses and adopted various mitigation mechanisms to ensure essential consumption. This study highlights the need for urban social policies that can support this vulnerable group amid a pandemic. We also discuss policy implications for cities and economic development with a focus on street vendors.

1. Introduction

In the urban areas of low- and middle-income countries, the nonagricultural informal sector has played and still plays a significant role. This sector is a major source of employment and earnings, constituting a large percentage of urban economies in developing countries (Bosch & Esteban-Pretel, 2012; Martínez et al., 2017). In urban areas, informal workers typically include the self-employed working at home, on the streets, or in the open air; salaried laborers in informal enterprises or their clients' homes; unpaid family workers; casual or day laborers (Chen, 2005, 2012; Hussmanns, 2004; Tienda & Raijman, 2000).

In developing cities worldwide, a large proportion of informal workers are street vendors, with the majority being women (Jensen & Peppard, 2003; Lemessa et al., 2021; Muyanja et al., 2011; Thanh et al., 2021; Williams & Gurtoo, 2012). Street vendors sell various products, such as food, drinks, and other small items (Chen, 2005; Jensen & Peppard, 2003; Lemessa et al., 2021; Romero-Michel et al., 2021; Sekhani et al., 2019). Since women street vendors are often characterized as underskilled immigrants with a low level of education, they represent the most disadvantaged and marginalized group (Boonjubun, 2017; Horn, 2010; Lemessa et al., 2021; Martínez et al., 2017; Muyanja et al., 2011; Thanh et al., 2021). Most vendors live in crowded semipermanent or unsettled houses that lack minimum sanitary conditions (McKay et al., 2016; Ullah et al., 2021). Also, they are highly vulnerable to socioeconomic crises (Balbuena & Skinner, 2020; Thanh et al., 2021). Additionally, since women vendors face educational and financial constraints (Pearce et al., 1988; Sisay et al., 2021), they tend to experience more adverse consequences and have fewer coping mechanisms during crises than male vendors (Cohen, 2010; Horn, 2010).

Unlike previous financial crises that directly affected the formal sector followed by indirectly affecting the informal sector (Cohen, 2010; Horn, 2010; Rani, 2020), the COVID-19 pandemic has directly affected informal sector workers such as street vendors (Rakshit & Basistha, 2020). To curb the pandemic, many countries have adopted social distancing or lockdown measures, which socioeconomically disrupt all economic sectors, including the informal sector (Chen, 2020; Laing, 2020; Rakshit & Basistha, 2020). Such disruptions have caused an economic burden for the street vendors, who lack sufficient skills, resources, coping strategies, and mitigation mechanisms (Balbuena & Skinner, 2020; Lemessa et al., 2021; Mustafa et al., 2021; Resnick, 2020; Sisay et al., 2021; Thai et al., 2021; Turner et al., 2021).

In developing cities, the urban poor and even middle-low-income

* Corresponding author. *E-mail addresses:* phamtienthanh@tdtu.edu.vn (P.T. Thanh), pbduong@vnua.edu.vn (P.B. Duong).

https://doi.org/10.1016/j.cities.2022.103879

Received 27 November 2020; Received in revised form 14 July 2022; Accepted 21 July 2022 Available online 28 July 2022 0264-2751/© 2022 Elsevier Ltd. All rights reserved. families rely on informal vending systems for daily necessities, for the convenience and the low prices (Boonjubun, 2017; Maneepong & Walsh, 2013; Martínez et al., 2017; McKay et al., 2016; Sekhani et al., 2019; Wertheim-Heck, 2020). Moreover, informal suppliers (e.g., farmers selling fresh produce to vendors) would lose their income if informal markets were to close (Dzawanda et al., 2021; Wegerif, 2020). Therefore, understanding the economic burden of the pandemic on street vendors is crucial, including their coping strategies and mitigation mechanisms for the sake of their livelihood and welfare, as well as those of other stakeholders (Dzawanda et al., 2021; Wegerif, 2020). With this in mind, this study seeks to understand the economic impacts of COVID-19 on street vendors in Vietnam by answering the following research questions:

- (i). How has the COVID-19 pandemic affected the women street vendors?
- (ii). Which groups among the women street vendors have been more affected?
- (iii). How have they sustained their business?
- (iv). How have they mitigated these adverse effects?

Previous studies have examined how COVID-19 social distancing hit the livelihoods of street vendors and how they sustained their businesses and ensured their essential consumption (e.g., housing and food) (Coletto et al., 2021; Guha et al., 2021; SEWA-Bharat, 2020; Thai et al., 2021; Toriro & Chirisa, 2021; Turner et al., 2021). However, no studies to date have provided a comprehensive picture of women street vendors during this challenging period. Our study contributes to literature and practice in several ways. First, it is among the early attempts to examine the economic burden that social distancing imposed on women street vendors, and the coping strategies and mitigation mechanisms that they adopted in response. Second, it carries an important message that no one should be left behind during a public health crisis, and that more resilient cities should be established in the long term.

The remainder of the paper is organized as follows. Section 2 focuses on the backgrounds of street vendors as seen across the globe as part of the informal urban economy. Section 3 provides an overview of the COVID-19 pandemic. Section 4 presents an analytical framework on the adverse effect of pandemic social distancing on street vendors. Section 5 outlines the research methodology. Section 6 provides the quantitative and qualitative results, with further discussion. Section 7 highlights the policy implications. Section 8 concludes the paper, discussing research limitations and future research directions.

2. Street vendors as part of the informal urban economy

A study by Hart (1973) was among the first attempts in the academic literature to coin the term "informal sector" in a Third World context (Portes et al., 1986; Portes & Sassen-Koob, 1987; Portes & Schauffler, 1993; Charmes, 2012; Truong, 2018; Truong et al., 2020). The informal economy refers to economic activities that occur beyond state regulation but exclude the supplying of illegal goods or services (Castells & Portes, 1989; Cross, 2000; Portes & Schauffler, 1993). Characteristics of the informal sector include ease of entry, family ownership, reliance on local or indigenous resources, small-scale but labor-intensive production, the use of adaptive technologies, under-skilling or lack of formal training, tax evasion, avoidance of labor codes, and unregistered status (Cross, 2000; Gërxhani, 2004). Accordingly, while the formal market is generally rule-based, structured, explicit, and predictable, the informal sector lacks these attributes or belongs to the poor (McFarlane & Waibel, 2016; Rigon et al., 2020). Such a dichotomous understanding considers formality and informality as distinct economic activities in the cities, which lead to assuming formality as the norm and informality as a problem needing a solution or correction (Acuto et al., 2019; Boonjubun, 2017; Cross, 2000; Rigon et al., 2020). A growing body of literature challenges this dichotomous view by debating whether informality and formality

represent a dualism or a continuum rather than a dichotomy (Banks et al., 2020; Guha-Khasnobis et al., 2006; Recio et al., 2017; Rigon et al., 2020).

The informal sector has played a significant role in employment creation in developing cities (Charmes, 2012; Rakshit & Basistha, 2020). For instance, according to recent studies, approximately 91 % of workers in India were active in the informal sector (Rakshit & Basistha, 2020), while the informal economy in Vietnam constituted a large source of employment (57.2–58.8 % of nonagricultural employment) (General Statistics Office of Vietnam (GSO) & International Labour Organization (ILO), 2018). Informal workers often experience unequal access to social protections, unstable employment, low income, long working hours, and no employment contracts (General Statistics Office of Vietnam (GSO) & International Labour Organization (ILO), 2018). They can have a wide range of jobs, with those working on the streets or in the open air without a stable or fixed place of work being the most visible (Chen, 2012; Truong, 2018).

In developing countries, street vending is an important element of the urban informal economy (Bhowmik, 2012; Cross, 2000). First, it plays a critical role in ensuring the basic needs (especially food) of the urban population. Since most products from street vendors are inexpensive when compared to those available in restaurants or shops (Bhowmik, 2005; Hiemstra et al., 2006; Maneepong & Walsh, 2013; Martínez et al., 2017; Wegerif, 2020), large segments of the urban population consume street products-especially low-income groups and those with no time to prepare meals (Hiemstra et al., 2006). Second, street vending can represent a survival strategy or a source of employment and income for poor, uneducated, low-skilled, and immigrant individuals (Bhowmik, 2005; Maneepong & Walsh, 2013; Sekhani et al., 2019; Truong, 2018). Street vendors also offer tourists a chance to experience local culture, and enjoy food and cuisine culture (Chuang et al., 2014; Huynh-Van et al., 2022). However, street vendors are generally labeled as informality, have or need no license to work, and evade taxes (Bell & Loukaitou-Sideris, 2014; Hummel, 2017; Lemessa et al., 2021; Martínez et al., 2017; Roever & Skinner, 2016). Their activities are often framed as an embodiment of traffic jams, unclean and unsafe streets, and a lack of hygienic and sanitary practices (Bhowmik, 2005; Huynh-Van et al., 2022; Muyanja et al., 2011).

The divergent schools of thought on informality and formality affect—and are manifested in—policies on urban informality, which includes street vending (Recio et al., 2017). According to modernist theory, urban planners believe that cities should be planned to be orderly and clean at all costs (Bell & Loukaitou-Sideris, 2014; Skinner, 2009; Skinner & Watson, 2020). Thus, street vending is considered a threat to cities' aesthetics and the residue of a pre-modern economy that is destined to disappear with modernity (Boonjubun, 2017; Chen, 2012; Cross, 2000; Portes & Sassen-Koob, 1987; Williams & Gurtoo, 2012). Accordingly, street vendors often experience the confiscation of goods and evictions from vending sites (Boonjubun, 2017; Chen, 2020; Lemessa et al., 2021; Onodugo et al., 2016). However, since street vending contributes greatly to food security and employment, its existence is inevitable, leading to debates in urban studies (Bell & Loukaitou-Sideris, 2014; Boonjubun, 2017; Onodugo et al., 2016).

3. The COVID-19 pandemic in the world and in Vietnam: An overview

The novel coronavirus disease (COVID-19) first appeared in China in mid-December 2019. In March 2020, the World Health Organization (WHO) declared COVID-19 a global pandemic. In addition to heavily damaging public health, the pandemic has caused socioeconomic disruptions due to social distancing measures (Kretzschmar et al., 2021; Laing, 2020; Rakshit & Basistha, 2020; Teslya et al., 2020).

Vietnam's large population and extensive trade with the rest of the world has made the country highly vulnerable during the COVID-19 pandemic. Therefore, the Vietnamese government was well-prepared for the battle against COVID-19 at a very early stage, even before detecting any cases of infection. After confirming the first COVID-19 cases in January 2020, the Vietnamese government implemented a range of preventive measures to contain the outbreak, with the most important being strict nationwide social distancing (Duong et al., 2020). In April 2020, the Vietnamese government mandated social distancing in most provinces and cities for three weeks. Thanks to these measures, Vietnam achieved remarkable success in containing the COVID-19 pandemic in the first pandemic year (Duong et al., 2020).

Despite its important role in containing the pandemic, social distancing is considered a draconian strategy that requires a substantial trade-off between public health and livelihood. It involves preventive measures such as restrictions on mass gatherings and travel, school shutdowns, the closure of non-essential public places, and sheltering or working at home—all of which have disrupted economic activities at the individual, firm, sector, and national levels in both the formal and informal economies. In particular, in Vietnam, as of June 2020, around 30.8 million people aged 15 years and older felt the negative effects of the COVID-19 pandemic, including those experiencing job loss, reduced working hours, income loss, staggered working hours, and rotating leave (General Statistics Office of Vietnam (GSO), 2020). Female, unskilled, migrant, and informal workers were among the most vulnerable groups during the pandemic (General Statistics Office of Vietnam (GSO), 2020; Nguyen et al., 2020).

During social distancing, some cities have allowed street vendors—especially those selling fresh fruits and vegetables—to operate; however, most developing cities did not allow street vendors to work (Chen, 2020; Guha et al., 2021; Lemessa et al., 2021; Sisay et al., 2021; Toriro & Chirisa, 2021). In urban Vietnam, the vendors were also required to suspend their businesses during social distancing (Thanh et al., 2021; Turner et al., 2021). Already a vulnerable group, the pandemic pushed their economic live into further distress.

Similar to elsewhere in the Global South, street vending activity serves an important role in urban Vietnam (Eidse et al., 2016; Huynh-Van et al., 2022). Also, since the Vietnamese government has aimed to ensure that no one is left behind during the pandemic (Duong et al., 2020; Thanh et al., 2021), it launched financial aid packages to support individuals affected by the pandemic—including street vendors (Nguyen et al., 2020; Prime Minister of Vietnamese Government, 2020). The street vendors also received various cash or in-kind support from charitable individuals, organizations, and landlords (Thai et al., 2021; Tung, 2020; Turner et al., 2021).

4. COVID-19 social distancing and street vendors: An analytical framework

To guide quantitative and qualitative data analysis, we formulated an analytical framework (see Fig. 1) that was adapted and modified from the livelihood framework in the context of vulnerability (Cannon et al., 2003; DFID, 1999; Serrat, 2017; Thanh & Duong, 2017). During COVID-19 social distancing, the following mechanisms may have led to adverse effects on street vendors.

First, these vendors experienced a reduction in purchasing power since their customers suffered from income shock, limited going out, and/or fear of infection from direct human contact (Kiaka et al., 2021; Nasution et al., 2021; Thanh et al., 2021). Second, vendors might have been restricted or prohibited from doing business (Chen, 2020; Guha et al., 2021: Gupte & Mitlin, 2021: Sisay et al., 2021: Toriro & Chirisa, 2021; Wegerif, 2020). For example, public places where vendors sold their goods were shut down, resulting in few or no customers (Nasution et al., 2021; Turner et al., 2021). Additionally, some vendors could not go to their selling locations due to public transportation shutdowns (Dzawanda et al., 2021; Resnick, 2020; Toriro & Chirisa, 2021). Third, street vending is generally a low-cost and simple business model, which makes it a competitive market with free entry (Maneepong & Walsh, 2013). Therefore, in a time of crisis like the COVID-19 pandemic, many formal workers have become unemployed, with some becoming street vendors as the last resort to maintain their livelihood (Allison et al., 2021; Maneepong & Walsh, 2013; Toriro & Chirisa, 2021). Accordingly, both old and new vendors could not avoid competing against one another for space and customers (Boonjubun, 2017; Maneepong & Walsh, 2013; Narayanan & Saha, 2021). Fourth, street vendors often lack technological skills and devices, as well as professional skills (Cohen, 2010; Maneepong & Walsh, 2013), which prevents them from adopting modern sales methods (e.g., online or phone sales).

When their businesses were heavily affected, vendors might adopt certain coping strategies to sustain their businesses or use mitigation mechanisms to ensure their essential consumption (Mustafa et al., 2021;



Fig. 1. Street vendors during COVID-19 pandemic: An analytical framework.

Thai et al., 2021; Thanh et al., 2021). Mitigation mechanisms and coping strategies refer to activities, measures, actions, or policies undertaken before or during the occurrence of negative events (e.g., disasters, emergencies, or crises) to avoid, minimize, or eliminate risks to individuals or property and reduce the actual or potential effects, severity, vulnerability, damage, hardship or suffering (de Siano et al., 2020; Shakeri et al., 2021). Preparing or activating mitigation mechanisms and coping strategies can occur at micro or macro levels (Wilhite, 2005). In this study, coping strategies and mitigation mechanisms refer to a wide range of actions and measures that street vendors have adopted in response to reductions in business and consumption.

During a crisis, street vendors could adopt various coping strategies to sustain their businesses, such as changing their products, services, or sales areas to reach new markets, or adopting modern sales methods and home delivery (Boonjubun, 2017; Coletto et al., 2021; Kiaka et al., 2021; Maneepong & Walsh, 2013; Nasution et al., 2021). In addition, the nature of their informality has made some vendors less likely to (frequently) follow social distancing measures, but they have used informal and often subtle tactics to navigate the surveillance of the local authorities and resist them in order to continue their businesses (Romero-Michel et al., 2021; Sisay et al., 2021; Thanh et al., 2021; Turner et al., 2021). For instance, in the Global South, vendors might work around local authorities' policing timetables and seek help from local residents to hide their goods during normal times and in crisis, with some playing "cat and mouse" or "hide and seek" with local authorities (Boonjubun, 2017; Kiaka et al., 2021; Lata et al., 2019; Thai et al., 2021; Turner et al., 2021).

Regarding mitigation mechanisms to ensure essential consumption during the crisis, some street vendors might prefer formal mechanisms when they are available, such as government relief, community-based support, savings, or formal loans (Martínez et al., 2021; Nguyen & Vu, 2020; Raju et al., 2021; Tung, 2020; Turner et al., 2021). However, many vendors were less able to access government support since the eligibility to obtain it might require complex documents and possibly complicated, time-consuming, and even unreasonable application procedures (Turner et al., 2021; Wertheim-Heck, 2020). For instance, since many vendors live in informal settlements with no official residence registration and often lack proof of income, they found it difficult to prove their eligibility for government support (Thai et al., 2021; Turner et al., 2021).

Due to insufficient or absent formal mitigation mechanisms, street vendors often resorted to informal coping mechanisms (e.g., alternative income from other family members, cash or in-kind support from friends or relatives, informal loans, or sales of assets) to cover their day-to-day expenses during the crisis (Martínez et al., 2021; Mustafa et al., 2021; Raju et al., 2021; Resnick, 2020; Turner et al., 2021). Since the COVID-19 pandemic has been neither a localized, trivial disruption nor an idiosyncratic shock but a global crisis, not only the vendors but also their family members and relatives experienced economic burdens (Resnick, 2020). Therefore, they might find it difficult to call on each other for assistance. Accordingly, some vendors borrowed from moneylenders at high interest rates, while some sold their durable assets in response to income shocks (Guha et al., 2021; Martínez et al., 2021; Turner et al., 2021). Consequently, with low repayment capacity and loss of productive assets, they may get into trouble or fall into (deeper) poverty over the long term (Agarwal, 2021; Guha et al., 2021; Turner et al., 2021).

5. Methodology

5.1. Study site and context

We conducted a cross-sectional survey and in-depth interviews with the street vendors in Ho Chi Minh City, the largest business hub in Vietnam, where street vending is an essential part of city life (Thanh et al., 2021). In this city, nonagricultural informal employment accounts for 51.6 % of total employment and 53.4 % of nonagricultural employment (Herrera et al., 2012). Street vendors account for 12 % of nonagricultural informal employment. Most street vendors are women, accounting for 66.7 % of total vendors (Herrera et al., 2012). In 2020, Ho Chi Minh City administratively comprised 5 rural districts and 19 urban districts. We selected five urban districts for this study (see Fig. 2). In Vietnam in general and Ho Chi Minh City in particular, street vending is considered a threat to urban modernization. Despite this, vendors are permitted to do business as long as they obey the regulations prescribed by the government (Prime Minister of Vietnamese Government, 2007).

Data were collected in June 2020. To contain the pandemic, Vietnam imposed strict nationwide social distancing from 1 April to 22 April 2020. After easing social distancing measures, Ho Chi Minh City entered the "new normal" period, and citizens could resume their daily activities and business. Thus, we asked respondents to recall and report information relevant to the three months preceding the survey and interviews.

5.2. Sampling process

In this study, a street vendor is defined as an informal petty trader offering goods and services for sale to the public on the street or in openair markets, without having a permanent built-up structure (Bhowmik, 2005). We asked women street vendors to participate in the survey and interviews. Recruitment was conducted through convenience sampling. Due to the pandemic, many people became unemployed, and then some became street vendors to eke out a living. Since we focused on how vendors' economic lives during the pandemic differed from the prepandemic period, we removed new-entrant vendors.

Prior to the official survey or interviews, a pilot survey and in-depth interviews with five women street vendors were conducted to ensure the instruments' logical consistency, wording, meaning, and appropriateness. Interviews occurred face-to-face with each respondent in the Vietnamese language (central dialect). The authors and two well-trained research assistants used extensive note-taking to collect quantitative and qualitative data, which was then translated into English for subsequent analysis.

We collected quantitative data using a structured questionnaire. A total of 100 women street vendors voluntarily participated in the survey. We removed the questionnaires of respondents who did not complete the survey or provided conflicting information. After data cleansing, the final sample for quantitative analysis included 91 respondents. To triangulate quantitative data, the authors and research assistants conducted in-depth interviews with 15 women street vendors to collect qualitative data. The interviews typically lasted 25–30 min each.

5.3. Empirical analysis

This study combined both quantitative and qualitative methods.

5.3.1. Quantitative analysis

Our survey questionnaire was adapted from various literature review, including empirical studies, and well-established surveys and reports (Companion, 2012; Hiemstra et al., 2006; Lin & (Lamson)., 2018; McKay et al., 2016; Roever & Skinner, 2016; Williams & Gurtoo, 2012). It was also based on our understanding of the reality of street vendors during COVID-19 social distancing. Accordingly, the final survey questions included the following sets of variables.

5.3.1.1. Economic burden. To answer the first and second research questions, we used reduced business and reduced daily consumption variables to measure the economic burden of COVID-19 social distancing on women vendors. The measures of these two indicators were developed from the Positive and Negative Affect Schedule (PANAS) (Watson et al., 1988), which is commonly used in social science to measure well-being (Brown & Ryan, 2003; Göb et al., 2007;



Fig. 2. Map of Ho Chi Minh City with surveyed districts.

Satici, 2016). During COVID-19 social distancing, some vendors (e.g., lottery-ticket vendors) completely stopped their vending when the Vietnamese government shut down all lottery-related activities (Nguyen & Vu, 2020). Similarly, some vendors selling nonessential products also suspended their business since their customers cut down spending on such products. Therefore, we constructed the reduced business variable using a modified scale with an additional response: "Completely affected (almost no sales)." Accordingly, we measured reduced business using a 6-point scale and reduced consumption using a 5-point scale, with higher values indicating more adverse effects (see Table 2). The analysis of both variables helped us comprehend the economic burden of COVID-19 social distancing on women vendors.

5.3.1.2. Coping strategies. To answer the third research question, we examined three coping strategies that are feasible and available to the women vendors during COVID-19 social distancing, including changing (i) commodities (e.g., from non-essential to essential goods), (ii) sales areas, and (iii) sales methods (e.g., online sales, then home delivery).

5.3.1.3. Mitigation mechanisms. To answer the fourth research question, we examined the various mitigation mechanisms that women vendors adopted to ensure their essential consumption (e.g., food and housing) during COVID-19 social distancing. The formal mechanisms included savings, support from government or charity programs, and formal loans. Meanwhile, informal mechanisms included support from friends, relatives, or landlords, reduced spending for nonessential goods and services, informal loans (e.g., from friends, relatives, or moneylenders), and sales of assets (i.e., nonproductive and productive assets).

5.3.1.4. Sociodemographic and vending characteristics. These consisted of two major sets of characteristics, including those of the street vendors (e.g., education, marital status, age, and immigration status) and their vending (e.g., experience, sales area, and commodity). These characteristics were used as explanatory variables in the ordered probit model (see Eq. 1).

Quantitative data obtained via the structured questionnaire were used for descriptive and inferential statistical analyses. Descriptive statistics examined the effects of COVID-19 social distancing on the business and consumption of the women vendors (first research question) and the coping strategies and mitigation mechanisms they adopted in response to these adverse effects (third and fourth research questions, respectively). Additionally, to answer the second research question—i. e., which groups among women vendors were more affected—we used quantitative data to estimate the determinants of their economic burden using inferential statistics. Given the ordinal nature (i.e., ordered responses) of outcome variables, a common method is to adopt an ordered probit (Cameron & Trivedi, 2005; Wooldridge, 2010). Thus, we estimated the following model:

$$y_i = X_i \beta + \varepsilon_i \quad (i = 1, 2, \dots, N) \tag{1}$$

where y_i is the self-reported economic burden of individual i, β represents the coefficients to be estimated, X_i denotes the vector of explanatory variables, and ε_i is the error term.

Based on Eq. 1 and the variables used for inferential analysis, we hypothesized that women vendors with more vending experience, better educated, married, or older ones were less affected than those with less vending experience, lower educated, unmarried (single, divorced, widowed), or younger ones, respectively. Immigrant vendors were hypothesized to be more affected than local ones, while those selling staple food or those vending in wet market, residential, or office areas were hypothesized to be less affected than those vending non-staple food or non-food items or selling in other areas, respectively. Positive and negative coefficients indicated greater and lesser effects, respectively.

5.3.2. Qualitative analysis

We used the qualitative data obtained from in-depth interviews with women vendors to examine the first, third, and fourth research questions. Accordingly, the interview questions focused on three themes: (i) how COVID-19 social distancing has affected their business and consumption; (ii) what coping strategies they have adopted to sustain their business; and (iii) what mitigation mechanisms they have adopted in response to reduced consumption. We analyzed and reported qualitative data thematically to triangulate the findings from descriptive statistics and capture a more comprehensive understanding. Following Braun and Clarke (2006), we first transcribed data and scrutinized the transcripts. Second, we identified and reviewed the themes. Third, we produced discussions associated with the research questions (Braun & Clarke, 2006).

6. Results and discussion

6.1. Sample profile

Table 1 shows that most respondents were married (89.0 %), with a

Table 1

Sample profile.

Variable(s)	Description	Ν	%		
Socio-demographic characteristics					
Marital status	Single, divorced, widowed	10	11.0 %		
	Currently married	81	89.0 %		
Education level	Never gone to school	14	15.4 %		
	Elementary school (1st–5th grade)	34	37.4 %		
	Secondary school (6th-9th grade)	31	34.1 %		
	High school (10th–12th grade)	12	13.2 %		
Age	Young adults (20-35 years old)	15	16.5 %		
	Middle-aged adults (36-55 years old)	54	59.3 %		
	Older adults (>55 years old)	22	24.2 %		
Immigration status	Local resident	22	24.2 %		
	Immigrant	69	75.8 %		
Vending characteristics					
Vending experience	\leq 2 year	23	25.3 %		
	2-<5 years	24	26.4 %		
	\geq 5 years	44	48.4 %		
Sales area	Schools, recreation centers	69	75.8 %		
	Residential or office areas	8	8.8 %		
	Wet market areas	14	15.4 %		
Commodity	Non-staple food or non-food items	66	72.5 %		
	Staple food	25	27.5 %		
Observations		91			

Source: Authors' calculation.

large proportion (59.3 %) comprising middle-aged adults, while around 24.2 % and 16.5 % were older and younger adults, respectively. These results are similar to findings from previous studies in which most vendors were married and middle-aged adults (Guha et al., 2021; Muyanja et al., 2011; Williams & Gurtoo, 2012). In line with previous studies (Guha et al., 2021; Martínez et al., 2017), many respondents were immigrants from other provinces and cities (75.8 %). In Vietnam, most young immigrants choose to work in factories for a better salary since employers typically favor younger workers. Accordingly, older immigrants tend to work as informal workers such as street vendors.

Some respondents (13.2 %) had completed high school, while many had completed elementary school (37.4 %) or secondary school (34.1 %), and some (15.4 %) never went to school. These findings were similar to other studies in which low educational attainment presumably characterized street vendors (Bhowmik, 2005; McKay et al., 2016; Muyanja et al., 2011; Romero-Michel et al., 2021). Many respondents (75.8 %) vended near schools or recreation centers, while the others vended in wet market areas (15.4 %) and near residential or office areas (8.8 %). Only around 27.5 % sold staple food (e.g., meat, fish, vegetables), while the remainder (72.5 %) sold non-staple food (e.g., beverages, snack food) and non-food items (e.g., lottery tickets, flowers). Nearly half (48.4 %) had vended on the streets for five years or longer, while 26.4 % vended for more than two years but less than five years,

Table 2

Economic burden during COVID-19 pandemic.

and 25.3 % vended for two years or less. These results align with other studies in which approximately half of the surveyed vendors had operated their businesses on the streets for five years or longer (Williams & Gurtoo, 2012). Our survey results revealed that the women street vendors shared similar characteristics to those across the world.

6.2. Economic burden

Table 2 reveals that COVID-19 social distancing caused a great economic burden among women vendors with respect to reduced business. Overall, around 12.1 % of respondents reported their business being slightly or moderately affected, 27.5 % reported being much or severely affected, and most (56 %) reported having almost no sales.

These results support the notion that the pandemic heavily affected women vendors (Abdul et al., 2021; Mustafa et al., 2021; Sisay et al., 2021; Turner et al., 2021). Some plausible reasons for the complete loss of business could be explained using the framework in Fig. 1. For instance, vendors were not allowed to operate during social distancing (Thai et al., 2021; Thanh et al., 2021). Moreover, some vendors (e.g., lottery-ticket vendors) had no product to sell since lottery-related activities were completely suspended (Nguyen & Vu, 2020). Table 2 also demonstrates that the vendors who suspended their businesses seemed to experience greater business loss.

However, an interesting result was that some vendors (4.4 %) reported their business being unaffected or even improving. For instance, according to a 57-year-old local vendor selling snack foods and fruits:

During the strict social distancing period due to the pandemic, my business was even better because my regular customers might get bored, and thus hanker for food. Currently, my business was even much lower than in that period.

Similar to other findings in the Global South (SEWA-Bharat, 2020; Thai et al., 2021; Turner et al., 2021), Table 2 also reveals that the COVID-19 pandemic adversely affected women vendors' consumption expenditure. In particular, many women vendors (69.2 %) reported being much or seriously affected, while the remainder (20.9 %) reported being slightly or moderately affected. Notably, a few vendors (9.9 %) reported that their consumption was unaffected, which perhaps indicates that their business was unaffected or that they had sufficient mitigation mechanisms to ensure their consumption.

The results from in-depth interviews showed that some vendors even experienced catastrophic burdens due to their own difficult circumstances or idiosyncratic shocks. For instance, some women vendors or their household members experienced sickness, injury, chronic diseases, or even hospitalization during this challenging time. Additionally, some vendors had family members working in the informal sector (e.g., motorbike taxi riders) who also suffered income loss during this period.

Variable(s)	Description	Whole sample		Continue vending		Stop vending	
		N	%	N	%	N	%
Reduced business	Reduction in business, measured by:						
(Mean = 4.97; SD = 1.41)	=1 if unaffected	4	4.4 %	4	11.8 %	0	0.0 %
	=2 if slightly affected	1	1.1 %	1	2.9 %	0	0.0 %
	=3 if moderately affected	10	11.0 %	7	20.6 %	3	5.3 %
	=4 if much affected	15	16.5 %	14	41.2 %	1	1.8 %
	=5 if seriously affected	10	11.0 %	7	20.6 %	3	5.3 %
	=6 if completely affected (almost no sales)	51	56.0 %	1	2.9 %	50	87.7 %
Reduced consumption	Reduction in daily consumption expenditure, measured by:						
(Mean = 3.70; SD = 1.31)	=1 if unaffected	9	9.9 %	7	20.6 %	2	3.5 %
	=2 if slightly affected	11	12.1 %	4	11.8 %	7	12.3 %
	=3 if moderately affected	8	8.8 %	3	8.8 %	5	8.8 %
	=4 if much affected	33	36.3 %	13	38.2 %	20	35.1 %
	=5 if seriously affected	30	33.0 %	7	20.6 %	23	40.4 %
Observations		91		34		57	

Source: Authors' calculation.

Accordingly, these unexpected idiosyncratic events, combined with adverse effects of the COVID-19 social distancing, caused them double or even triple shocks (Dzawanda et al., 2021). For instance, according to a 72-year-old immigrant vendor selling beverages:

My husband could not work because of his sickness. My son worked very little, while my daughter-in-law left home. My family had three young grandchildren... When I vended on the street, our grandchildren took care of one another (the oldest one was seven years old)... One day (in the time of social distancing), I was vending when I got high blood pressure; thus, I was admitted to hospital.

Similarly, according to a 35-year-old immigrant vendor selling beverages:

In a rough time due to the pandemic, I suspended my business, and my husband was also unable to do his job as a motorbike taxi rider. We did not have money for consumption. My children and my husband were sick, and my father-in-law was weak. Economic life was difficult. I had to shoulder the responsibility of the whole family.

Table 2 also unveils the noteworthy finding that many vendors (37.4 %) continued to conduct their business. Regardless of the high risk of being caught and fined, they did not (frequently) follow the social distancing measures since no work meant hunger or deprivation (Coletto et al., 2021; Dzawanda et al., 2021; Majithia, 2020; Romero-Michel et al., 2021; Thanh et al., 2021). Due to their informality, they used various informal and often subtle tactics to maintain their vending activities and resist the local authorities (Romero-Michel et al., 2021; Turner et al., 2021). Most women vendors were hard-hit, especially those experiencing other non-pandemic problems or suspending their businesses, which underscores the need for social policies targeting this vulnerable group during a pandemic.

6.3. Determinants of economic burden

The severity of COVID-19 social distancing impacts on women vendors might depend on their socio-demographics and vending specifics (Chen, 2020; Gupte & Mitlin, 2021; Kiaka et al., 2021; Nasution et al., 2021; SEWA-Bharat, 2020; Wertheim-Heck, 2020). Table 3 shows that most explanatory variables were statistically insignificant, indicating that the economic burden of the pandemic did not differ among the women vendors of varying marital status, age, education, vending experience, or commodity group. Since the social distancing measures have wreaked havoc on the economy by causing direct and indirect detrimental effects on street vending activity, women vendors experienced similar business loss and reduced consumption regardless of their sociodemographic characteristics.

On the contrary, immigrant vendors experienced greater business losses but no difference in reduced consumption when compared to local vendors. This result is similar to findings in Southeast Asia and India, where migrant vendors experienced a greater burden during lockdowns (Sripriya, 2021; Turner et al., 2021). A plausible reason is that immigrants typically face social-capital constraints and political marginalization (e.g., fewer regular customers or lack of connections with the local authority) that limit their ability to sustain their business during social distancing. However, immigrants might receive more food aid from sociopolitical or community-based organizations that have considered them a more vulnerable group (Thai et al., 2021; Tung, 2020; Turner et al., 2021). Thus, they could save some money from food purchases to spend on non-food necessities. Additionally, some migrants might have received rent reductions or exemptions from their landlords during the pandemic (Thai et al., 2021). This support could explain why immigrant vendors experienced more business loss than local vendors, yet no difference in reduced consumption.

Regarding the sales areas of vendors, those vending near schools or recreation centers incurred no difference in economic burden when compared to those vending near residential or office areas; however,

Table 3

Determinants of the economic burden during COVID-19 pandemic.

Variable(s)	Reduced business		Reduced consumption			
	Model 1	Model 2	Model 3	Model 4		
Socio-demographic characteristics of street vendors						
Marital status (Ref = Currently married)	0.669*	0.404	-0.503	-0.741		
3	(0.086)	(0.315)	(0.375)	(0.165)		
Education level (Ref = Never go to school)						
Elementary school	-0.186	-0.427	-0.478	-0.785		
,	(0.591)	(0.267)	(0.190)	(0.055)		
Secondary school	0.101	-0.099	-0.426	-0.723		
5	(0.798)	(0.809)	(0.288)	(0.070)		
High school	0.735	0.138	-0.089	-0.695		
0	(0.155)	(0.793)	(0.852)	(0.187)		
Age (Ref = Young adults)						
Middle-aged adults	-0.157	-0.499	-0.372	-0.555		
U	(0.752)	(0.327)	(0.326)	(0.243)		
Older adults	0.153	-0.337	-0.470	-0.899		
	(0.789)	(0.588)	(0.273)	(0.086)		
Immigration status (Ref =	0.597**	0.776***	0.023	0.207		
Local resident)	(0.026)	(0.008)	(0.934)	(0.505)		
Characteristics of street vending	g (01020)	(01000)	(01501)	(0.000)		
Vending experience (Ref $= \le 2$ years)	5					
2-<5 years		0.269		0.259		
		(0.485)		(0.476)		
\geq 5 years		0.132		0.195		
		(0.730)		(0.546)		
Sales area (Ref = Schools,						
recreation centers)						
Residential or office areas		-0.030		0.222		
		(0.946)		(0.605)		
Wet markets		-0.953**		-1.006**		
		(0.016)		(0.018)		
Commodity (Ref = Staple		0.308		-0.310		
food)						
		(0.357)		(0.248)		
Constant cut 1	-0.746	-1.430**	-2.444***	-3.179***		
	(0.198)	(0.012)	(0.001)	(0.000)		
Constant cut 2	-0.647	-1.320**	-1.920***	-2.591***		
	(0.275)	(0.019)	(0.006)	(0.000)		
Constant cut 3	-0.021	-0.658	-1.644 **	-2.296***		
	(0.973)	(0.288)	(0.018)	(0.002)		
Constant cut 4	0.559	-0.053	-0.647	-1.251*		
	(0.359)	(0.933)	(0.332)	(0.081)		
Constant cut 5	0.878	0.284				
	(0.149)	(0.654)				
Observations	91	91	91	91		

Source: Authors' calculation.

Note: robust p-value in parentheses.

 ** and *** represent statistical significance at the 5 % and 1 % levels, respectively.

those vending in wet market areas were less adversely affected. Similar to the case of Indonesia (Nasution et al., 2021), lockdowns shut schools and recreation centers and restricted office areas; therefore, students had to adopt online learning, office staff often had to work from home, and people limited their gatherings and going out. Office staff not working from home might also avoid vendors for fear of contracting the coronavirus (Nasution et al., 2021). Furthermore, under the strict measures for minimizing community infection risk, the street vendors selling in office areas had to suspend their activities, which severely affected their business. Since wet market areas provided essential and indispensable goods, trading activities took place almost as usual or with little restriction. In wet market areas, trading activities were under strict surveillance and public health measures were also implemented by the local authorities; therefore, the women vendors in these areas could still operate, and thus their businesses were less adversely affected. Our results demonstrated almost no difference in economic burden among women vendors with socio-demographic differences, except for

migration status and vending location.

6.4. Coping strategies to sustain business

The analytical framework (see Fig. 1) implied that the adverse consequences of the COVID-19 pandemic on the business of street vendors would be more serious if they had insufficient or no coping strategies. Table 4 reveals that most respondents (94.5 %) had no strategy to sustain their business. Moreover, around 4.4 % did not resort to any coping strategy because their business was unaffected. Only one vendor changed her sales area, while none changed their commodities or sales methods. Our results are inconsistent with previous studies in Indonesia and some African countries, where the street vendors changed their products, moved to other locations, or adopted online sales during the pandemic (Kiaka et al., 2021; Nasution et al., 2021; Toriro & Chirisa, 2021).

We found that almost no women vendors had coping strategies. This could be due to various reasons associated with street vending specifics during both the normal state and the pandemic. First, due to the nature of their jobs-i.e., simple production and inflexible business (Maneepong & Walsh, 2013)—they were less likely to adapt to turbulent and uncertain events. Second, the vendors tended to sell products that took advantage of their expertise (e.g., processed food) and the availability of affordable wholesalers or suppliers of raw materials. Therefore, they lacked the flexibility to change to other products. Third, they often worked in the specific areas with which they were familiar in order to maintain their existing networks and regular customers and reduce their operating costs (Bell & Loukaitou-Sideris, 2014; Milgram, 2014; Onodugo et al., 2016). They were also not allowed to move freely to sell their products during the pandemic (Dzawanda et al., 2021). Last, their low educational attainment and lack of technological skills prevented them from adopting modern sales methods. Most vendors were found to lack coping strategies, which underscores the importance of support programs to help them sustain their businesses during times of crisis.

6.5. Mitigation mechanisms in response to reduced consumption expenditure

Table 5 shows that around 31.9 % received at least one cash or inkind support, to ensure their daily consumption. In particular, approximately 11 % received support from friends or relatives, while around 9.9 % benefited from charity programs, and around 5.5 % received help from their landlords (e.g., reduced or exempted rent). Therefore, similar to elsewhere in the Global South (Martínez et al., 2021; Raju et al., 2021), the women vendors in Vietnam received support from various sources, and their social networks played an important role in supporting them during a crisis. Although the Vietnamese government released a \$2.7-billion financial aid package for the poor and firms affected by COVID-19 (Prime Minister of Vietnamese Government, 2020), and the street vendors were eligible to receive this support, only a small proportion (15.4 %) reported receiving this benefit, which is in line with findings in Colombia (Martínez et al., 2021). Similar to the cases in Southeast Asia and India (Raju et al., 2021; Thai et al., 2021; Turner et al., 2021), many women vendors in Vietnam did not meet the requirements for such support because they lacked proof of residence in

Table 4

Coping strategies to sustain business.

Strategies	Ν	%
Not applicable due to being unaffected	4	4.4 %
Change sales area	1	1.1 %
Change commodity	0	0.0 %
Change sales method	0	0.0 %
No solution	86	94.5 %
Observations	91	

Source: Authors' calculation.

Table 5

Mitigation mechanisms in response to reduced consumption expenditure.

Mechanisms	Ν	%
Savings	25	27.5 %
Cash or in-kind supports	29	31.9 %
From friends or relatives	10	11.0 %
From local authorities or government	14	15.4 %
From charity programs	9	9.9 %
From landlords	5	5.5 %
Income from other family members	22	24.2 %
Reduced expenditure for non-essential goods and services	77	84.6 %
Borrowings	40	44.0 %
From friends or relatives	23	25.3 %
From moneylenders at high interest	22	24.2 %
From financial institutions	1	1.1 %
Sales of assets	18	19.8 %
Non-productive assets	17	18.7 %
Productive assets	1	1.1 %
Observations	91	

Source: Authors' calculation.

the city or proof of having an income below the poverty line, while eligible vendors also faced difficulties in obtaining this financial aid due to complicated paperwork and procedures. For instance, according to a 72-year-old immigrant vendor selling beverages:

Many of my neighbors went back and forth many times to apply for the support, which was complicated. I was too weak, so I could not travel too much to receive the support.

When the supports were unavailable or insufficient, the women vendors resorted to other mechanisms to ensure their consumption. In particular, around 27.5 % used savings, which is similar to the result in Colombia (Martínez et al., 2021). Meanwhile, 24.2 % used money from other family members whose income was affected less or not at all. Additionally, a large proportion (84.6 %) reported reducing non-essential consumption, which is much different from the finding in Colombia (only 29 % reported this type of reduction) (Martínez et al., 2021). However, since the vendors tended to adopt a minimalist life-style, their nonessential consumption often accounted for a relatively small proportion of their total spending; therefore, they could not save much money by reducing nonessential consumption to spend on essential items.

Another common mitigation mechanism that the women vendors in Vietnam adopted was to borrow money, which is similar to other findings in the Global South (Guha et al., 2021; Martínez et al., 2021; Turner et al., 2021). Around 44 % reported having at least one loan. In particular, only one vendor reported having access to formal loans. Due to their low educational attainment and lack of assets for collateral, they were less likely to access formal loans (McKay et al., 2016; Mustafa et al., 2021). Approximately 25.3 % and 24.2 % borrowed from friends or relatives and high interest-charging moneylenders, respectively, which was also in line with the case in Colombia (Martínez et al., 2021). This raised the important concern that vendors still resorted to obtaining high-interest loans from moneylenders. Since they might have a low repayment capacity due to business loss, they might be pushed into (deeper) debt and poverty, resulting in their daily life potentially becoming chaotic (Guha et al., 2021). For instance, according to a 55year-old immigrant vendor selling snack foods, beverages, and cigarettes:

My daughter could not repay her debt and thus had to flee to evade her creditors. I must take care of my grandchild... I also had to borrow from moneylenders at high interest to re-invest in my business after the social distancing period.

Similarly, according to a 45-year-old immigrant vendor selling snack foods:

To be honest, I fled to evade my creditors in recent months. I had just re-opened my business for a few days.

Sales of assets was also considered a last resort for the women vendors, with around 19.8 % reporting selling their assets. Our results are in line with other findings in the Global South (SEWA-Bharat, 2020; Turner et al., 2021). Similar to high-interest loans from moneylenders, sales of assets might push them into more trouble and potentially mark a slide into poverty (Agarwal, 2021; Guha et al., 2021; Turner et al., 2021).

Overall, our results revealed that only some women vendors could access formal mitigation mechanisms, while many adopted various informal mechanisms (see Table 5). Nonetheless, most women vendors still suffered a large reduction in consumption (*see* Table 2), indicating that the mechanisms were insufficient. Therefore, this study highlights the importance of supporting programs to ensure that the basic needs of this vulnerable group are met during a pandemic.

7. Policy implications

Although street vending plays an important role in developing cities, this activity has been viewed as an inhibition to urban modernization, resulting in debates regarding whether street vending should be banned from cities (Boonjubun, 2017; Onodugo et al., 2016; Sekhani et al., 2019; Yatmo, 2008). As such, the policy implications associated with street vending can be a complex puzzle. This study highlights some policy implications that could help to sustain and improve the short- and long-term livelihoods and welfare of street vendors during a public health crisis and under normal conditions.

During a pandemic, street vendors in densely populated cities might be at high risk of becoming infected with and spreading disease to their communities. However, vendors have been stuck in a real dilemma during the COVID-19 pandemic: to die from starvation or the virus (Chen, 2020). Therefore, many continued their businesses regardless of the high risk. This study highlights the need for urban policies that focus on supporting vendors and reducing contagion risk to this group during a pandemic by calling for various policy responses and the participation of various stakeholders. In the short run, first, the vendors may only be willing to suspend their business only when they can ensure their basic needs. Therefore, they should be provided with food assistance, housing subsidies, and cash grants (Moroz et al., 2020). However, vendors might not have their registration of temporary residence in a city or proof of income below the poverty line, which makes them ineligible for government supports. Thus, the government should streamline the application procedure and increase the effectiveness and outreach of relief or aid to vendors (Raju et al., 2021; Turner et al., 2021). Second, vendors should receive health safety guidelines, masks, sanitizers or soaps, and a map of water points (Majithia, 2020; Skinner & Watson, 2021). Such guidelines and supports would help street vendors ensure compliance with health and safety measures, thereby reducing the risk of disease infection and spread. Additionally, since many urban populations rely on street food, vendors selling staple food should be supported in continuing their business. As long as vendors follow the appropriate disease safety guidelines and prevention measures, they can trade as safely as supermarkets (Skinner & Watson, 2021; Wegerif, 2020). To enhance the effectiveness of these activities, the adoption of information technology should be fostered to quickly and easily identify and support eligible beneficiaries. The roles of various stakeholders, such as the Women's Union, the Youth Union, the Red Cross, press agencies, and landlords, should be promoted to connect donors and charity organizations, collect information from vendors, deliver safety guidelines and supports to the vendors, and supervise their compliance with public health safety measures. Furthermore, local officials should also play a proactive role in guiding and monitoring street vendors to ensure that the vendors strictly follow safety guidelines while also obeying current government regulations related to street vending activity.

In the longer term, urban agendas should include the formalization

and integration of informal vendors into urban life (Bell & Loukaitou-Sideris, 2014; Song, 2020). Thereafter, they could continue their business and even improve their income while the urban authorities could better manage vending activities and effectively support this group in both normal and crisis states (Martínez et al., 2017; Song, 2020). When formalized, street vendors may: (i) be required to register with different local authorities for a working permit or business license; (ii) incur higher costs (e.g., taxes, license fees); (iii) be allowed to only vend at designated locations and times; or (iv) be required to follow some regulations (e.g., hygienic and sanitary) (Bhowmik, 2003; Hummel, 2017; Kregor, 2017; Lemessa et al., 2021; Roever, 2006). Due to such complications and requirements, not all vendors are willing to (fully) formalize their businesses (Bell & Loukaitou-Sideris, 2014; Bhowmik, 2003; Prasetyo, 2021). Therefore, the formalization of street vending would require great efforts, a multi-action plan, and the participation of various stakeholders-including the street vendors themselves. To this end, the government should first conduct surveys on whether (and the extent to which) vendors are willing to formalize or comply with regulations, as well as the supports they need to do so. Then, bottom-up policies can be designed and combined with top-down policies to enhance the feasibility and effectiveness of policies in practice. Second, communication work should focus on raising vendors' awareness so that they can view formalization as their right to better and more sustainable livelihoods, as well as their duty for the sake of modern and safe cities. Third, the vendors should also receive formal training on hygiene, sanitation, and entrepreneurial, professional, and technological skills (Maneepong & Walsh, 2013; Song, 2020). Fourth, more designated places for street vending activities should be planned, constructed, and equipped with basic facilities to enhance food safety, hygiene, and cleanliness (Ojeda & Pino, 2019; Song, 2020). Last but not least, urban development plans must consider street vendors as part of the planning process (Bhowmik, 2003; Onodugo et al., 2016).

8. Conclusions, limitations, and future research directions

8.1. Conclusions

This study examined the economic life of women street vendors during the COVID-19 pandemic. Our analysis revealed that they were heavily affected by the pandemic. This was particularly true for immigrant vendors and those selling near schools or recreation centers since they experienced a greater economic burden than their counterparts. Almost all vendors lacked coping strategies to sustain their businesses. Thus, they resorted to various mitigation mechanisms to ensure their consumption. Since they lacked access to formal mitigation mechanisms, they had to adopt various informal ones; however, such mitigation seemed insufficient.

This study makes some significant contributions to the literature and practice. First, it used mixed methods to triangulate data and consolidate findings on the economic burden and resilience among women street vendors during COVID-19 social distancing. Second, it examined various topics and indicators associated with economic burden and resilience. Third, it conceptualized an analytical framework of adverse effects of the pandemic on women vendors in the presence or absence of coping strategies and mitigation mechanisms. Finally, it proposed urban social and development policies for developing countries that aim to sustain the livelihoods of vendors and protect public health during a pandemic. Other long-term policy implications for cities and economic development, with a focus on street vendors, were also discussed. Our policy implications are important not only for the ongoing pandemic but also for similar crises in the future. This study carries the important message that no one-especially vulnerable groups-should be left behind and that more resilient cities should be established. Thus, Sustainable Development Goals (e.g., SDG1, SDG2, and SDG3) can be achieved or sustained.

8.2. Limitations and future study directions

Despite its contributions, this study is not without certain limitations. First, its use of a small sample and use of convenience sampling may have created some statistical concerns and limited generalizability. As such, future studies should include a larger sample and recruit respondents using random or stratified sampling. Another methodological limitation was the reliance on subjective scales to measure the economic burden of the pandemic. Second, this study did not examine the vendors' perceptions regarding the effectiveness of mitigation mechanisms. Thus, future studies should consider this issue to capture a deeper understanding of their hardships during the pandemic, even when they adopted some mitigation mechanisms. Third, our study implied that street vending should be formalized so that the vendors can receive more benefits or supports in both the normal state and in a crisis. However, the formalization of this activity may be associated with more complicated requirements and higher costs; thus, not all vendors would be willing to formalize their business (Bell & Loukaitou-Sideris, 2014; Bhowmik, 2003; Prasetyo, 2021). Future studies should focus on their willingness to formalize and the support they need to do so, which can provide more detailed, practical, and feasible policies.

Availability of data and material

The datasets generated during and/or analyzed during the current study are available from the corresponding author on reasonable request.

CRediT authorship contribution statement

Pham Tien Thanh: Data curation, Formal analysis, Investigation, Methodology, Project administration, Software, Supervision, Visualization, Writing – original draft. **Pham Bao Duong:** Funding acquisition, Resources, Validation, Writing – review & editing.

Declaration of competing interest

No potential conflict of interest is reported by the authors.

Data availability

Data will be made available on request.

Acknowledgements

We are grateful to the editors and the anonymous reviewers for their valuable and insightful comments. We wish to thank our research assistants, especially Ms. Le Thi Bao Ngan, for their support during the field works.

References

- Abdul, A., Negi, D. P., Rani, A., & Kumar, S. (2021). The impact of COVID-19 on migrant women workers in India. Eurasian Geography and Economics, 62(1), 93–112. https:// doi.org/10.1080/15387216.2020.1843513
- Acuto, M., Dinardi, C., & Marx, C. (2019). Transcending (in)formal urbanism. Urban Studies, 56(3), 475–487. https://doi.org/10.1177/0042098018810602
- Agarwal, B. (2021). Imperatives of recognising the complexities: Gendered impacts and responses to COVID-19 in India. *Economia Politica*, 1–23. https://doi.org/10.1007/ S40888-021-00242-8/TABLES/1
- Allison, N., Ray, K., & Rohel, J. (2021). Mobilizing the streets: The role of food vendors in urban life. Food, Culture & Society, 24(1), 2–15. https://doi.org/10.1080/ 15528014.2020.1860454
- Balbuena, P., & Skinner, C. (2020, June 7). For world's street vendors, life may never be the same after COVID-19. Retrieved August 31, 2020, from https://www.wiego.org/blo g/worlds-street-vendors-life-may-never-be-same-after-covid-19.
- Banks, N., Lombard, M., & Mitlin, D. (2020). Urban informality as a site of critical analysis. The Journal of Development Studies, 56(2), 223–238. https://doi.org/ 10.1080/00220388.2019.1577384

- Bell, J. S., & Loukaitou-Sideris, A. (2014). Sidewalk informality: An examination of street vending regulation in China. *International Planning Studies*, 19(3–4), 221–243. https://doi.org/10.1080/13563475.2014.880333
- Bhowmik, S. K. (2003). National policy for street vendors. Retrieved from Economic and Political Weekly, 38(16) https://www.jstor.org/stable/4413453.
- Bhowmik, S. K. (2005). Street vendors in Asia : A review. *Economic and Political Weekly*, 40(22/23), 2256–2264, 4416705.
- Bhowmik, S. K. (2012). Street vendors in the global urban economy. Taylor & Francis. Boonjubun, C. (2017). Conflicts over streets: The eviction of Bangkok street vendors. *Cities*, 70, 22–31. https://doi.org/10.1016/j.cities.2017.06.007
- Bosch, M., & Esteban-Pretel, J. (2012). Job creation and job destruction in the presence of informal markets. *Journal of Development Economics*, 98(2), 270–286. https://doi. org/10.1016/j.jdeveco.2011.08.004

Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. Qualitative Research in Psychology, 3(2), 77–101. https://doi.org/10.1191/1478088706qp0630a

- Brown, K. W., & Ryan, R. M. (2003). The benefits of being present: Mindfulness and its role in psychological well-being. Retrieved from *Journal of Personality and Social Psychology*, 84(4), 822 https://psycnet.apa.org/buy/2003-02410-012.
- Cameron, A. C., & Trivedi, P. K. (2005). Microeconometrics: Methods and applications. Cambridge, UK: Cambridge University Press.
- Cannon, T., Twigg, J., & Rowell, J. (2003). Social vulnerability, sustainable livelihoods and disasters. London: Department for International Development (DFID).
- Castells, M., & Portes, A. (1989). World underneath: The origins, dynamics, and effects of the informal economy. In A. Portes, M. Castells, & L. A. Benton (Eds.), *The informal economy: Studies in advanced and less developed countries* (pp. 11–37). Baltimore: Johns Hopkins University Press.
- Charmes, J. (2012). The informal economy worldwide: Trends and characteristics. Margin: The Journal of Applied Economic Research, 6(2), 103–132. https://doi.org/ 10.1177/097380101200600202
- Chen, M. (2005). Rethinking the informal economy: Linkages with the formal economy and the formal regulatory environment. Hanoi. Retrieved from. Helsinki: WIDER Research Paper 2005/10 http://hdl.handle.net/10419/63329.
- Chen, M. (2012). The Informal Economy: Definitions, Theories and Policies (WIEGO Working Paper No. 1). Retrieved from https://www.wiego.org/sites/default/files/publicat ions/files/Chen_WIEGO_WP1.pdf.
- Chen, M. (2020). COVID-19, cities and urban informal workers: India in comparative perspective. The Indian Journal of Labour Economics, 63(S1), 41–46. https://doi.org/ 10.1007/s41027-020-00254-1
- Chuang, Y. F., Hwang, S. N., Wong, J. Y., & Chen, C. D. (2014). The attractiveness of tourist night markets in Taiwan – A supply-side view. *International Journal of Culture*, *Tourism, and Hospitality Research*, 8(3), 333–344. https://doi.org/10.1108/IJCTHR-09-2013-0067
- Cohen, J. (2010). How the global economic crisis reaches marginalised workers: The case of street traders in Johannesburg. South Africa. Gender & Development, 18(2), 277–289. https://doi.org/10.1080/13552074.2010.491345
- Coletto, D., Jaber, L., & Vanhellemont, L. (2021). Street vendors in Lima in the time of COVID-19: Guilty or oppressed? *The Canadian Geographer / Le Géographe Canadien*, 65(4), 435–447. https://doi.org/10.1111/cag.12712
- Companion, M. (2012). Urban and peri-urban cultivation in northern Mozambique: Impacts on food security among female street food vendors. *Journal of Applied Social Science*, 6(2), 149–164. https://doi.org/10.1177/193672441244428
- Cross, J. (2000). Street vendors, and postmodernity: Conflict and compromise in the global economy. *International Journal of Sociology and Social Policy*, 20(1/2), 29–51. https://doi.org/10.1108/01443330010789061
- DFID. (1999). Sustainable livelihoods guidance sheets. London: The Department for International Development.
- Duong, D. M., Le, V. T., & Ha, B. T. T. (2020). Controlling the COVID-19 pandemic in Vietnam: Lessons from a limited resource country. Asia Pacific Journal of Public Health, 32(4), 161–162. https://doi.org/10.1177/1010539520927290
- Dzawanda, B., Matsa, M., & Nicolau, M. (2021). Poverty on the rise: The impact of the COVID-19 lockdown on the informal sector of GweruZimbabwe. *International Social Science Journal*, 71(S1), 81–96. https://doi.org/10.1111/issj.12285
- Eidse, N., Turner, S., & Oswin, N. (2016). Contesting street spaces in a socialist city: Itinerant vending-scapes and the everyday politics of mobility in HanoiVietnam. Annals of the American Association of Geographers, 106(2), 340–349. https://doi.org/ 10.1080/00045608.2015.1117936
- General Statistics Office of Vietnam (GSO). (2020). Report on the impact of the COVID-19 on labour and employment situation in Viet Nam. Retrieved from https://www. gso.gov.vn/en/data-and-statistics/2020/07/report-on-the-impact-of-the-covid-19on-labour-and-employment-situation-in-viet-nam/.
- General Statistics Office of Vietnam (GSO), International Labour Organization (ILO). (2018). The 2016 Report on Informal Employment in Viet Nam. Hanoi. Retrieved from https://www.ilo.org/hanoi/Whatwedo/Publications/WCMS_638330/lang_en/ index.htm.
- Gërxhani, K. (2004). The informal sector in developed and less developed countries: A literature survey. *Public Choice*, 120(3/4), 267–300. https://doi.org/10.1023/B: PUCH.0000044287.88147.5e
- Göb, R., McCollin, C., & Ramalhoto, M. F. (2007). Ordinal methodology in the analysis of likert scales. Quality & Quantity, 41(5), 601–626. https://doi.org/10.1007/s11135-007-9089-z
- Guha, P., Neti, A., & Lobo, R. (2021). Merging the public and private spheres of women's work: Narratives from women street food vendors during Covid-19 crisis. *Gender, Work & Organization, 1–17.* https://doi.org/10.1111/gwao.12772
- Guha-Khasnobis, B., Kanbur, R., & Ostrom, E. (2006). Linking the formal and informal economy. New York: Oxford University Press. https://doi.org/10.1093/ 0199204764.001.0001

Gupte, J., & Mitlin, D. (2021). COVID-19: What is not being addressed. *Environment and Urbanization*, 33(1), 211–228. https://doi.org/10.1177/0956247820963961

Hart, K. (1973). Informal income opportunities and urban employment in Ghana. Retrieved from *The Journal of Modern African Studies*, 11(1), 61–89 https://www.jsto r.org/stable/159873.

- Herrera, J., Kuépié, M., Nordman, C. J., Oudin, X., & Roubaud, F. (2012). Informal sector and informal employment: Overview of data for 11 cities in 10 developing countries (No. 9). Cambridge, MA. Retrieved from. In Women in informal employment: Globalizing and organizing (WIEGO) https://www.wiego.org/publications/informalsector-and-informal-employment-overview-data-11-cities-10-developing-countries.
- Hiemstra, A. M. F., van der Kooy, K. G., & Frese, M. (2006). Entrepreneurship in the street food sector of Vietnam - assessment of psychological success and failure factors. *Journal of Small Business Management*, 44(3), 474–481. https://doi.org/ 10.1111/j.1540-627X.2006.00183.x
- Horn, Z. E. (2010). The effects of the global economic crisis on women in the informal economy: Research findings from WIEGO and the inclusive cities partners. *Gender & Development*, 18(2), 263–276. https://doi.org/10.1080/13552074.2010.491339

Hummel, C. (2017). Disobedient markets: Street vendors, enforcement, and state intervention in collective action. *Comparative Political Studies*, 50(11), 1524–1555. https://doi.org/10.1177/0010414016679177

Hussmanns, R. (2004). Defining and measuring informal employment. Geneva: International Labour Organization.

- Huynh-Van, B., Vuong-Thao, V., Huynh-Thi-Thanh, T., Dang-Xuan, S., Huynh-Van, T., Tran-To, L.Nguyen-Viet, H., ... (2022). Factors associated with food safety compliance among street food vendors in Can Tho city, Vietnam: Implications for intervention activity design and implementation. *BMC Public Health*, 22(1), 1–11. https://doi.org/10.1186/S12889-022-12497-2/TABLES/4
- Ivic, S. (2020). Vietnam's response to the COVID-19 outbreak. Asian Bioethics Review, 12 (3), 341–347. https://doi.org/10.1007/s41649-020-00134-2
- Jensen, R., & Peppard, D. M. (2003). Hanoi's informal sector and the Vietnamese economy: A case study of roving street vendors. *Journal of Asian and African Studies*, 38(1), 71–84. https://doi.org/10.1177/002190960303800104
- Kiaka, R., Chikulo, S., Slootheer, S., & Hebinck, P. (2021). "The street is ours". A comparative analysis of street trading, Covid-19 and new street geographies in Harare, Zimbabwe and Kisumu, Kenya. *Food Security*, 13(5), 1263–1281. https://doi. org/10.1007/s12571-021-01162-y
- Kregor, E. (2017). How do you formalize a tamale: How to ease street vendors' transition out of the shadow economy. UC Irvine Law Review, 7, 453.

Kretzschmar, M. E., Rozhnova, G., & van Boven, M. (2021). Isolation and contact tracing can tip the scale to containment of COVID-19 in populations with social distancing. *Frontiers in Physics*, 8, 677. https://doi.org/10.3389/fphy.2020.622485

- Laing, T. (2020). The economic impact of the coronavirus 2019 (Covid-2019): Implications for the mining industry. *The Extractive Industries and Society*, 7(2), 580–582. https://doi.org/10.1016/j.exis.2020.04.003
- Lata, L., Walters, P., & Roitman, S. (2019). A marriage of convenience: Street vendors' everyday accommodation of power in Dhaka, Bangladesh. *Cities*, 84, 143–150. https://doi.org/10.1016/j.cities.2018.08.002
- Lemessa, S. D., Watabaji, M. D., Yismaw, M. A., & Tadesse, A. B. (2021). Evening street vending and the tragic living conditions of vendors: The case of eastern Ethiopia region. *Cities*, 108, Article 102947. https://doi.org/10.1016/j.cities.2020.102947
- Lin, S., & (Lamson).. (2018). We work like ants...We avoid being troublemaker. International Journal of Sociology and Social Policy, 38(11/12), 1024–1040. https:// doi.org/10.1108/IJSSP-01-2018-0008
- Majithia, A. S. (2020, July). Impact of COVID-19 on street vendors in India: Status and steps for advocacy. Retrieved from https://www.wiego.org/impact-covid-19-street-ven dors-india-status-and-steps-advocacy.
- Maneepong, C., & Walsh, J. C. (2013). A new generation of Bangkok street vendors: Economic crisis as opportunity and threat. *Cities*, 34, 37–43. https://doi.org/ 10.1016/j.cities.2012.11.002
- Martínez, L., Short, J. R., & Estrada, D. (2017). The urban informal economy: Street vendors in Cali, Colombia. *Cities*, 66, 34–43. https://doi.org/10.1016/j. cities.2017.03.010
- Martínez, L., Young, G., Trofimoff, V., Valencia, I., Vidal, N., Espada, A. D., & Roblesa, E. (2021). The hardships of the poorest during the COVID-19 pandemic: Data about the socioeconomic conditions and governance of informal workers. *Data in Brief, 40*, Article 107728. https://doi.org/10.1016/j.dib.2021.107728

McFarlane, C., & Waibel, M. (2016). Urban informalities: Reflections on the formal and informal. London: Routledge.

- McKay, F. H., Singh, A., Singh, S., Good, S., & Osborne, R. H. (2016). Street vendors in Patna, India: Understanding the socio-economic profile, livelihood and hygiene practices. *Food Control*, 70, 281–285. https://doi.org/10.1016/j. foodcont.2016.05.061
- Milgram, B. L. (2014). Remapping the edge: Informality and legality in the Harrison road night market, Baguio CityPhilippines. *City and Society*, 26(2), 153–174. https://doi. org/10.1111/ciso.12038
- Moroz, H., Shrestha, M., & Testaverde, M. (2020). Potential responses to the COVID-19 outbreak in support of migrant workers. Washington, DC: World Bank. https://doi.org/ 10.1596/33625
- Mustafa, F., Khursheed, A., Fatima, M., & Rao, M. (2021). Exploring the impact of COVID-19 pandemic on women entrepreneurs in Pakistan. *International Journal of Gender and Entrepreneurship*, 13(2), 187–203. https://doi.org/10.1108/LJGE-09-2020-0149/FULL/PDF
- Muyanja, C., Nayiga, L., Brenda, N., & Nasinyama, G. (2011). Practices, knowledge and risk factors of street food vendors in Uganda. *Food Control*, 22(10), 1551–1558. https://doi.org/10.1016/j.foodcont.2011.01.016

- Narayanan, S., & Saha, S. (2021). Urban food markets and the COVID-19 lockdown in India. *Global Food Security*, 29, Article 100515. https://doi.org/10.1016/j. gfs.2021.100515
- Nasution, N., Sarmini, S., Warsono, W., Wasino, W., & Shintasiwi, F. (2021). Using coping strategies of informal sector traders amid COVID-19 in Indonesia. Retrieved from *Journal of Social Studies Education Research*, 12(3), 144–174 https://www.le arntechlib.org/p/219957/.
- Nguyen, H. T. T., Nguyen, T. T., Dam, V. A. T., Nguyen, L. H., Vu, G. T., Nguyen, H. L. T., & Le, H. T. (2020). COVID-19 employment crisis in Vietnam: Global issue, national solutions. *Frontiers in Public Health, 8*, Article 590074. https://doi.org/10.3389/ fnubb.2020.590074
- Nguyen, T. H. D., & Vu, D. C. (2020). Impacts of the COVID-19 pandemic upon mental health: Perspectives from Vietnam. Psychological Trauma: Theory, Research, Practice, and Policy, 12(5), 480–481. https://doi.org/10.1037/tra0000694
- Ojeda, L., & Pino, A. (2019). Spatiality of street vendors and sociospatial disputes over public space: The case of ValparaísoChile. *Cities*, 95(February), Article 102275. https://doi.org/10.1016/j.cities.2019.02.005
- Onodugo, V. A., Ezeadichie, N. H., Onwuneme, C. A., & Anosike, A. E. (2016). The dilemma of managing the challenges of street vending in public spaces: The case of Enugu City, Nigeria. *Cities*, 59, 95–101. https://doi.org/10.1016/j. cities 2016 06.001
- Pearce, T. O., Kujore, O. O., & Agboh-Bankole, V. A. (1988). Generating an income in the urban environment: the experience of street food vendors in Ile-Ife, Nigeria. *Africa*, 58(4), 385–400. https://doi.org/10.2307/1160348
- Portes, A., & Sassen-Koob, S. (1987). Making it underground: Comparative material on the informal sector in Western market economies. *American Journal of Sociology*, 93 (1), 30–61. https://doi.org/10.1086/228705
- Portes, A., & Schauffler, R. (1993). Competing perspectives on the Latin American informal sector. Population and Development Review, 19(1), 33. https://doi.org/ 10.2307/2938384
- Portes, A., Blitzer, S., & Curtis, J. (1986). The urban informal sector in Uruguay: Its internal structure, characteristics, and effects. World Development, 14(6), 727–741. https://doi.org/10.1016/0305-750X(86)90015-X
- Prasetyo, E. H. (2021). Legitimacy building of digital platforms in the informal economy: Evidence from Indonesia. *Journal of Entrepreneurship in Emerging Economies*. https:// doi.org/10.1108/JEEE-02-2021-0073. ahead-of-p(ahead-of-print).
- Prime Minister of Vietnamese Government. (2007, April). Decree No. 39/2007/ND-CP on individuals doing trade independently and regularly do not have to register for their businesses.
- Prime Minister of Vietnamese Government. (2020, April 9). The resolution 42/NQ-CP 2020 assistance for people affected by Covid-19 pandemic. Retrieved September 12, 2020, from https://tulieuvankien.dangcongsan.vn/Uploads/2020/4/7/17/NQ-42. signed.pdf.
- Raju, E., Dutta, A., & Ayeb-Karlsson, S. (2021). COVID-19 in India: Who are we leaving behind? Progress in Disaster Science, 10, Article 100163. https://doi.org/10.1016/j. pdisas.2021.100163
- Rakshit, B., & Basistha, D. (2020). Can India stay immune enough to combat COVID-19 pandemic? An economic query. *Journal of Public Affairs*, 20(4), Article e2157. https://doi.org/10.1002/pa.2157
- Rani, U. (2020). Old and new forms of informal employment. In M. Chen, & F. Carré (Eds.), *The informal economy revisited: Examining the past, envisioning the future* (1st ed., pp. 88–91). London: Routledge. https://doi.org/10.4324/9780429200724.
- Recio, R. B., Mateo-Babiano, I., & Roitman, S. (2017). Revisiting policy epistemologies on urban informality: Towards a post-dualist view. *Cities*, 61, 136–143. https://doi.org/ 10.1016/j.cities.2016.08.018
- Resnick, D. (2020). COVID-19 lockdowns threaten Africa's vital informal urban food trade. In J. Swinnen, & J. McDermott (Eds.), COVID-19 and global food security (pp. 73–74). Washington, DC: International Food Policy Research Institute (IFPRI). https://doi.org/10.2499/p15738coll2.133762.
- Rigon, A., Walker, J., & Koroma, B. (2020). Beyond formal and informal: Understanding urban informalities from Freetown. *Cities*, 105, Article 102848. https://doi.org/ 10.1016/j.cities.2020.102848
- Roever, S. (2006). Enforcement and compliance in Lima's street markets: The origins and consequences of policy incoherence towards informal traders. In B. Guha-Khasnobis, R. Kanbur, & E. Ostrom (Eds.), *Linking the formal and informal economy*. New York: Oxford University Press. https://doi.org/10.1093/0199204764.001.0001.
- Roever, S., & Skinner, C. (2016). Street vendors and cities. Environment and Urbanization, 28(2), 359–374. https://doi.org/10.1177/0956247816653898
- Romero-Michel, J. C., Mokay-Ramírez, K. A., Delgado-Machuca, M., Delgado-Enciso, J., Aurelien-Cabezas, N. S., Tiburcio-Jimenez, D.Delgado-Enciso, I., ... (2021). Health and economic measures in response to the COVID-19 pandemic-effect on street vendors. *The Journal of Infection in Developing Countries*, 15(02), 198–203. https:// doi.org/10.3855/jidc.13465
- Satici, S. A. (2016). Psychological vulnerability, resilience, and subjective well-being: The mediating role of hope. *Personality and Individual Differences*, 102, 68–73. https://doi.org/10.1016/j.paid.2016.06.057
- Sekhani, R., Mohan, D., & Medipally, S. (2019). Street vending in urban 'informal' markets: Reflections from case-studies of street vendors in Delhi (India) and Phnom Penh City (Cambodia). *Cities, 89*, 120–129. https://doi.org/10.1016/j. cities.2019.01.010
- Serrat, O. (2017). The sustainable livelihoods approach. In *Knowledge solutions* (pp. 21–26). Singapore: Springer Singapore. https://doi.org/10.1007/978-981-10-0983-9 5.
- SEWA-Bharat. (2020). Gendered precarity in the lockdown. Retrieved from. In All India federation of self-employed women's association https://www.sewabharatresearch. org/wp-content/uploads/2020/05/Gendered_Precarity_SB_Lockdown-1.pdf.

- Shakeri, E., Vizvari, B., & Nazerian, R. (2021). Comparative analysis of disaster management between India and Nigeria. *International Journal of Disaster Risk Reduction*, 63, Article 102448. https://doi.org/10.1016/J.IJDRR.2021.102448
- de Siano, R., Sciabolazza, V. L., & Sapio, A. (2020). Economic resilience and regional disparities: The value added of spatial analysis. In R. De Siano, V. L. Sciabolazza, & A. Sapio (Eds.), *Regional resilience to climate and environmental shocks* (pp. 7–29). Cham: Springer. https://doi.org/10.1007/978-3-030-545888-8.2
- Sisay, M., Gashaw, T., Degefu, N., Hagos, B., Alemu, A., Teshome, Z.Dessie, Y., ... (2021). One in five street traditional coffee vendors suffered from depression during the COVID-19 pandemic in Harar town, Ethiopia. *Neuropsychiatric Disease and Treatment*, 17, 2173–2182. https://doi.org/10.2147/NDT.S315370
- Skinner, C. (2009). The struggle for the streets: Processes of exclusion and inclusion of street traders in DurbanSouth Africa. *Development Southern Africa*, 25(2), 227–242. https://doi.org/10.1080/03768350802090709
- Skinner, C., & Watson, V. (2020). The informal economy in urban Africa: Challenging planning theory and praxis. In M. Chen, & F. Carré (Eds.), *The informal economy revisited: Examining the past, envisioning the future* (1st ed., pp. 123–131). London: Routledge. https://doi.org/10.4324/9780429200724.
- Skinner, C., & Watson, V. (2021). Planning and informal food traders under COVID-19: The South African case. *Town Planning Review*, 92(3), 301–307. https://doi.org/ 10.3828/tpr.2020.38
- Song, S. (2020). Street stall economy in China in the post-COVID-19 era: Dilemmas and regulatory suggestions. *Research in Globalization*, 2(1575), Article 100030. https:// doi.org/10.1016/j.resglo.2020.100030
- Sripriya, S. S. (2021). Lives of migrants during COVID-19 pandemic: A study of mobile vendors in Mumbai. In R. Akhtar (Ed.), *Coronavirus (COVID-19) outbreaks*, *environment and human behaviour* (pp. 143–155). Cham, Switzerland: Springer. https://doi.org/10.1007/978-3-030-68120-3_9.
- Teslya, A., Pham, T. M., Godijk, N. G., Kretzschmar, M. E., Bootsma, M. C. J., & Rozhnova, G. (2020). Impact of self-imposed prevention measures and short-term government-imposed social distancing on mitigating and delaying a COVID-19 epidemic: A modelling study. *PLoS Medicine*, 17(7), Article e1003166. https://doi. org/10.1371/journal.pmed.1003166
- Thai, H. M. H., Dinh, P. Q., & Nguyen, P. T. (2021). The resilience of street vendors in surviving the COVID-19 crisis in Hanoi, Vietnam. In R. van Melik, P. Filion, & B. Doucet (Eds.), Volume 3: Public space and mobility (1st ed., pp. 155–164). Bristol University Press. https://doi.org/10.2307/j.ctv1t4m1m6.21.
- Thanh, P. T., & Duong, P. B. (2017). Health shocks and the mitigating role of microcredit—The case of rural households in Vietnam. *Economic Analysis and Policy*, 56, 135–147. https://doi.org/10.1016/j.eap.2017.08.006
- Thanh, P. T., Nguyen, H. T. H., Ngan, L. T. B., Nguyen, D. M. D., Phan, G. H., & Nguyen, T. M. N. (2021). No one left behind: Risk communication to the street vendors during COVID-19 social distancing. *Library Hi Tech*, 40(2), 357–375. https://doi.org/10.1108/LHT-03-2021-0100

- Tienda, M., & Raijman, R. (2000). Immigrants' income packaging and invisible labor force activity. Retrieved from Social Science Quarterly, 81(1), 291–310 https://www. jstor.org/stable/42864383.
- Toriro, P., & Chirisa, I. (2021). Vendors on wheels! The changing terrain and manifestation of informality in Harare under Covid-19 pandemic restrictions. *Cogent Social Sciences*, 7(1), 1–11. https://doi.org/10.1080/23311886.2021.1939230
- Truong, V. D. (2018). Tourism, poverty alleviation, and the informal economy: The street vendors of HanoiVietnam. *Tourism Recreation Research*, 43(1), 52–67. https://doi. org/10.1080/02508281.2017.1370568
- Truong, V. D., Liu, X., & Pham, Q. (2020). To be or not to be formal? Rickshaw drivers' perspectives on tourism and poverty. *Journal of Sustainable Tourism*, 28(1), 33–50. https://doi.org/10.1080/09669582.2019.1665056
- Tung, L. T. (2020). Social work responses for vulnerable people during the COVID-19 pandemic: The role of socio-political organisations. Asia Pacific Journal of Social Work and Development, 1–8. https://doi.org/10.1080/02185385.2020.1837216
- Turner, S., Langill, C. J., & Nguyen, N. B. (2021). The utterly unforeseen livelihood shock: COVID-19 and street vendor coping mechanisms in Hanoi, Chiang Mai and Luang Prabang. Singapore Journal of Tropical Geography, 42(3), 484–504. https://doi. org/10.1111/sjtg.12396
- Ullah, A. A., Nawaz, F., & Chattoraj, D. (2021). Locked up under lockdown: The COVID-19 pandemic and the migrant population. Social Sciences & Humanities Open, 3(1), Article 100126. https://doi.org/10.1016/j.ssaho.2021.100126
- Watson, D., Clark, L. A., & Tellegen, A. (1988). Development and validation of brief measures of positive and negative affect: The PANAS scales. Retrieved from *Journal* of *Personality and Social Psychology*, 54(6), 1063 https://psycnet.apa.org/buy/1988-31508-001.
- Wegerif, M. C. A. A. (2020). "Informal" food traders and food security: Experiences from the Covid-19 response in South Africa. *Food Security*, 12(4), 797–800. https://doi. org/10.1007/s12571-020-01078-z
- Wertheim-Heck, S. (2020). April 8). The impact of the COVID-19 lockdown on the diets of Hanoi's urban poor. In *International Institute for Environment and Development*.
- Wilhite, D. A. (2005). Drought policy and preparedness: The Australian experience in an international context. In L. C. Botterill, & D. A. Wilhite (Eds.), From disaster response to risk management (pp. 157–176). Dordrecht: Springer. https://doi.org/10.1007/1-4020-3124-6_12.
- Williams, C. C., & Gurtoo, A. (2012). Evaluating competing theories of street entrepreneurship: Some lessons from a study of street vendors in BangaloreIndia. *International Entrepreneurship and Management Journal*, 8(4), 391–409. https://doi. org/10.1007/s11365-012-0227-2
- Wooldridge, J. M. (2010). Econometric analysis of cross section and panel data. Cambridge MA: MIT Press.
- Yatmo, Y. A. (2008). Street vendors as "out of place" urban elements. Journal of Urban Design, 13(3), 387–402. https://doi.org/10.1080/13574800802320889